Author’s response to reviews

Title: Staphylococcus aureus with an erm-mediated constitutive macrolide-lincosamide streptogramin B resistance phenotype has reduced susceptibility to the new ketolide, solithromycin

Authors:

Weiming Yao (yao719082506@126.com)
Guangjian Xu (1355125180@qq.com)
Duoyun Li (liduoyun94@163.com)
Bing Bai (568646535@qq.com)
Hongyan Wang (doctorwanghongyan@163.com)
Hang Cheng (473806126@qq.com)
Jinxin Zheng (710194599@qq.com)
Xiang Sun (1006231877@qq.com)
Zhiwei Lin (linzw8406@hotmail.com)
Qiwen Deng ( qiwendeng@hotmail.com)
Zhijian Yu (yuzhijiansmu@163.com)

Version: 3 Date: 27 Jan 2019

Author’s response to reviews:

To: Dear Dr. Robin L. Cassady-Cain

Re: Manuscript ID: INFD-D-18-01517R2

On behalf of all authors, I am very grateful to you and the reviewers for the instructive comments and suggestions on improving our manuscript “Staphylococcus aureus with an erm-mediated constitutive macrolide-lincosamide streptogramin B resistance phenotype has reduced susceptibility to the new ketolide, solithromycin”.

We have revised the manuscript accordingly and answered questions and comments point by point as follows.
The newly revised manuscript contains 261 words for Abstract, 2635 words for Text (not including Abstract, Declarations, References and Tables), two tables and three figures in the main body; and three supplementary tables and one supplementary figure. The revised texts are highlighted in blue. We hope that our newly revised manuscript have answered your questions. And we appreciate your kind consideration and instructive comments.

Thanks again for your time and favorable consideration.

Best Regards,

Zhijian YU, MD
Qiwen Deng, MD

E-mail: yuzhijiansmu@163.com; qiwendeng@hotmail.com
Telephone: 0086775-26553111-26402
Fax: 0086755-26565025